Author's response to reviews

Title: A synergistic interaction between transcription factors nuclear factor-kappaB and signal transducers and activators of transcription 3 promotes gastric cancer cell migration and invasion

Authors:

Jiyeon Yoon (ghdghdghd00@daum.net)
Sung Jin Cho (chohan0504@hanmail.net)
Young San Ko (77ys77@daum.net)
Jinju Park (ydejuni@naver.com)
In Chan Hwang (in.hwang@mail.mcgill.ca)
Sang Yeun Han (seh229@cornell.edu)
Seon Young Nam (everest@snu.ac.kr)
Min A Kim (meesooch@snu.ac.kr)
Mee Soo Chang (mw9195@hanmail.net)
Hye Seung Lee (seonynam@khnp.co.kr)
Woo Ho Kim (woohokim@snu.ac.kr)
Byung Lan Lee (dslanat@snu.ac.kr)
Dong Hoon Shin (drdoogi@snu.ac.kr)

Version: 3 Date: 25 January 2013

Author's response to reviews: see over
Dr. Tonilynn Manibo  
on behalf of Dr. Wayne Phillips  
BMC Gastroenterology

January 25, 2013

Dear Dr. Manibo,

We are very happy to hear a good news from you,  
and highly appreciate your kind explanation of the revision format.

We have changed the followings as recommended:  
1. We changed “expressions” to “expression” (page 11, 9 lines from bottom)  
2. We changed “epithelila-mesenchymal” to “epithelial-mesenchymal” (page 13, line 5)  
3. We changed “expressions” to “expression” (page 13, line 17)  
4. We changed “confirme” to “confirm” and “expressi ons” to “expression” (page 14, line 9)  
5. We changed “expressions” to “expression” (page 15, 6 lines from bottom)  
6. We changed “activations” to “activation” (page 17, line 6)

Thank you again for your advice.

Please let us know if there any more things we need to change further.

Yours sincerely,

Byung Lan Lee, MD, PhD  
Professor, Department of Anatomy  
Seoul National University College of Medicine  
103 Daehak-ro, Jongno-gu, Seoul 110-799, Korea  
Phone: +82 2 740 8218; Fax: +82 2 745 9528  
E-mail: dslanat@snu.ac.kr